Author’s response to reviews

Title: Early identification and preventive care for elevated cardiovascular disease risk within a remote Australian Aboriginal primary health care service.

Authors:

Christopher P Burgess (paul.burgess@menzies.edu.au)
Ross S Bailie (ross.bailie@menzies.edu.au)
Christine M Connors (Christine.Connors@nt.gov.au)
Richard D Chenhall (chenhall@unimelb.edu.au)
Robyn A McDermott (robyn.mcdermott@unisa.edu.au)
Kerin O'Dea (kerin.odea@unisa.edu.au)
Charlie Gunabarra (Charlie.Gunabarra@nt.gov.au)
Hellen L Matthews (Hellen.Matthews@nt.gov.au)
Adrian J Esterman (adrian.esterman@unisa.edu.au)

Version: 5 Date: 26 January 2011

Author’s response to reviews:

The ‘final proof’ manuscript you sent me is still in the word document form and is not a publication proof. The manuscript still includes my directions to the production team about where tables and figures should be placed.

The Y axis labels in Figures 2 & 3 have been omitted during the production process. This makes the figures difficult to understand and changes them significantly from the versions that were approved during the peer review process. Please reinstate the Y axis labels and scaling - as approved by the peer reviewers.